ENDCAP C MEETING

DATE:

Wednesday 31st August 2005

TIME:

11.00 am ( Coffee from 10.30 a.m. )

VENUE
            Room 337, Oliver Lodge Laboratory

  DRAFT MINUTES

This meeting was the first meeting specifically dealing with Endcap C and following on from the series of UK-V Meetings. In Ian’s absence the agenda was compiled by Craig and Neil. 

Ian retires this year and I am sure you will all join me in thanking him for all his good work in organising the UK-V Meetings over the past years. The agendas were all circulated in good time, the presentations were made available on the WEB and the minutes were circulated promptly. Best wishes Ian in his retirement and I am sure he will still follow the progress of ATLAS when data starts rolling in.

All the presentations for this meeting may be found at

http://hep.ph.liv.ac.uk/~jnj/Endcap_C_31-8-05/
The minutes will be brief and mainly concerned with actions.

1. Minutes from Liverpool meeting
2. Actions from the Liverpool meeting

The minutes were accepted and the actions had taken place with the exception of the plan for shift working. This is still on-going as we are constrained by the deliveries of the LMTs

3. Review of module status, re-work and shipments
       Scott Lindsay 10min

ACTION
Surplus modules to be sent to NIKHEF

Scott

4. Report on disc testing



       Paul Dervan   20min

ACTION
Complete analysis of results




Paul

ACTION
Ensure all relevant data is uploaded to the database Joost/Paul/Joe

ACTION
Use same mapping for RODs and PSs in SR1
Jo/Paul/Joost

5. Database for use in SR1



      Joe Foster
       10min

Joe was unable to attend so the discussion took place in the last item.

6. Thermal disc studies 



       Tim Jones       15min

ACTION
Continue analysis



Tim/ Joost+students

7. Report on optical fibre tests



  Tim Brodbeck    10min

A good deal of discussion here. The attenuation seen is due to light in the cladding whereas light in the core suffers a trivial amount of attenuation. The effect of optical grease seems to be significant.

ACTION
Check on availability of infineons at CERN
Jo / Tim B

8. Mounting services onto the cylinder                                                          

Experience gained, timing, problems etc

Richard French    20min

A lot of discussion about the problems highlighted by Richard.

ACTION
Develop solution to mating of connectors at PPF1 and TPPF1C

ACTION
Decide who is responsible for mounting the 4 DCS panels  Jo/Craig

ACTION
Look into interference of the support ring for the thermal 

                         shield with the capilliaries and exhaust pipes

Richard

9. Warm testing of disc 9



    Scott Lindsay    20min

ACTION
Write up the procedure



Scott

10. Leak testing of cooling pipes on cylinder and                                        testing the temporary cooling circuits with D1
     Tim Jones
       10min

This is work in progress

11. Current planning and schedule


Craig Buttar
       20min

Craig updated the schedule but it is all dependent on deliveries

12. Preparations in SR1including a discussion                                                                of what needs to be sent to CERN


Jo Pater
       15min

ACTION
The Liverpool DCS system should act as back-up

Jo/Joost

ACTION
Look at preparing the thermal enclosure asap in SR1    Jo /Rich

13. Transport arrangements


         Stephen Haywood   10min

In progress.

14. Progress with transport frame


Peter Cooke          10min 

In progress.

ACTION
Develop a dummy load for the first test run
Peter / Richard

15. AOB

Next Meeting

Wednesday November 9th at Liverpool

Neil Jackson 1/9/2005

